Author's response to reviews

Title: Social Networks and Implementation of Evidence-Based Practices in Public Youth-Serving Systems: A Mixed Methods Study

Authors:

Lawrence A Palinkas (palinkas@usc.edu)
Ian W Holloway (ihollowa@usc.edu)
Eric Rice (ericr@usc.edu)
Dahlia Fuentes (dfuentes@usc.edu)
Qiaobing Wu (qiaobing.wu@gmail.com)
Patricia Chamberlain (pattic@oslc.org)

Version: 3 Date: 2 August 2011

Author's response to reviews: see over
August 2, 2011

RE: MS #1708727357516378 “Social Networks and Implementation of Evidence-Based Practices in Public Youth-Serving Systems: A Mixed Methods Study”

Dear Implementation Science Editorial Team:

Attached please find a revised version of the above referenced manuscript. In accordance with the comments and suggestions provided by the editors, we have reduced the length of the manuscript and formatted the manuscript in accordance with editorial guidelines. The word length of the text is now 5,742.

Please let us know at your earliest convenience if there is anything else you require from us.

Sincerely yours,

Lawrence A. Palinkas, Ph.D.
Albert G. and Frances Lomas Feldman Professor of Social Policy and Health